Author's response to reviews

Title: Demand for long acting and permanent contraceptive methods and associated factors among married women of reproductive age group in Debre Markos Town, North West Ethiopia.

Authors:

Gizachew Abdissa Bulto Mr. (gizachab@yahoo.com)
Tatek Abate Zeleke Mr. (tatekonline@gmail.com)
Teresa Kisi Beyen Mr. (terek7@gmail.com)

Version: 5 Date: 26 February 2014

Author's response to reviews: see over
To: Editorial Board  
BMC Women’s Health  

February 26, 2014

Dear sir /madam

It is to be recalled that we have sent a paper entitled ‘Demand for long acting and permanent contraceptive methods and associated factors among married women of reproductive age group in Debre Markos Town, North West Ethiopia’ to be published on your journal. Thus, as per the reviewer’s comments we have modified and made some language corrections of the document one by one.

Thank you for your consideration!